Author's response to reviews

Title: Effects of Ginsenoside Re on LPS-induced Inflammatory Mediators in BV2 Microglial Cell

Authors:

Kang-Woo Lee (kwlee@kiom.re.kr)
So Young Jung (yzzim@kiom.re.kr)
Sun-Mi Choi (choism@kiom.re.kr)
Eun Jin Yang (yej4823@hanmail.net)

Version: 3 Date: 17 October 2012

Author's response to reviews: see over
Dear Editor-in-Chief of BMC Complementary & Alternative Medicine

Ref: MS 1053585881765365

Title: Effects of Ginsenoside Re on LPS-induced Inflammatory Mediators in BV2 Microglial Cell

On the behalf of my co-authors, I would like to thank you for arranging the re-review of our manuscript and for your invitation to submit a revised version. We appreciate the effort of the reviewers, and believe that their constructive suggestions have resulted in a stronger manuscript for readers of the BMC Complementary & Alternative Medicine.

Yours faithfully,

Eun Jin Yang, PhD
Dear Reviewer,

RE: Effects of Ginsenoside Re on LPS-induced Inflammatory Mediators in BV2 Microglial Cell (MS: 1053585881765365)

We appreciate the minor comment from the reviewer regarding our manuscripts. According to his (her) suggestions, my colleagues and I have made the appropriate changes in a point by point manner. The responses are detailed as follows:

Reviewer(s)' Comments to Author:

**Comments of Reviewer #1:**

1. In the revised manuscript the authors highlighted the parts that they revised and replyed my comments in the cover letter. However, I would like to see the reply for the concern on the dose range in the Discussion but not only the cover letter.

➔ **Answers and revised points**

According to the reviewer’s comment, we added the reply for the concern on the dose range in the Discussion of a manuscript and marked the change in yellow.